Author's response to reviews

Title: Study Design of a Randomized Clinical Trial of an Internet-Based Adolescent Depression Preventive Intervention

Authors:

Tracy G Gladstone (tgladsto@wellesley.edu)
Monika Marko (mmarko@uic.edu)
Phyllis Rothberg (phyllisrothberg@gmail.com)
Jennifer Nidetz (jnidetz@uic.edu)
Anne Diehl (adiehl.contact@gmail.com)
Daniela T DeFrino (ddfefrino@uic.edu)
Mary Harris (mharris@wellesley.edu)
Eumene Ching (eumene_ching@vmed.org)
Milton Eder (milton.eder@accesscommunityhealth.net)
Jason Canel (JCanel@northshore.org)
Carl Bell (cbell@uic.edu)
William Beardslee (william.beardslee@childrens.harvard.edu)
C. Hendricks Brown (hendricks.brown@northwestern.edu)
Kathleen Griffiths (kathy.griffiths@anu.edu.au)
Benjamin Van Voorhees (bvanvoor@uic.edu)

Version: 2 Date: 13 January 2015

Author's response to reviews:

Dear Editors,

Here are the details of our revision:

Title page is p. 1 and includes the title, authors, author affiliations, author emails and corresponding author details

Abstract is p.2 and does not exceed 350 words. It follows the CONSORT extension for abstracts. The abstract has separate sections headed as Background, Methods/Design, Discussion. There are no abbreviations. The date of registration with the trial registration number is at the end of the abstract.

There is a title for the background section.

The conclusion and limitations setions are incorporated into the discussion section.

The names of all ethical bodies that approved the study are in the methods section.

Thank you,
Authors